\begin{table}%t1
\caption{\label{tab.KnownRCB.GenInfo}Known confirmed and candidate Magellanic RCB stars located in the EROS-2 fields; coordinates from 2MASS.}
\par
%%\centerline

\par
{\small \begin{tabular}{llcll}
\hline \hline
Star name & EROS-2 star & Coordinates ($J_{2000}$) & Other identifier & Large decline\\
 & identifier & from 2MASS & & seen by EROS-2?\\
\hline
& & Confirmed RCB & & \\
& & or DYPer stars & & \\
\object{HV 5637}$^{a}$ & \object{lm0181l22619} & 05:11:31.37 --67:55:50.6 & KDM 2492 & no\\
& & & MACHO-20.5036.12 \\
\object{W Men}$^{a}$ & \object{lm0584k5872} & 05:26:24.52 --71:11:11.8 & MACHO-21.7407.7 & yes \\
\object{HV 12842}$^{a}$ &  & 05:45:02.87 -64:24:22.8 & &\\
\object{MACHO-11.8632.2507}$^{b}$ & \object{lm0027m19359} & 05:33:48.94 --70:13:23.4 & HV~2671 & yes  \\
\object{MACHO-81.8394.1358}$^{b}$ & \object{lm0025k33192} & 05:32:13.36 --69:55:57.8 & & yes  \\
\object{MACHO--6.6575.13}$^{c}$ & \object{lm0016k29468} & 05:20:48.21 --70:12:12.5 & & yes, remained faint most of the time\\
\object{MACHO-6.6696.60}$^{c}$ & \object{lm0016m10260} & 05:21:47.98 --70:09:56.9 & HV 942 & yes \\
\object{MACHO-12.10803.56}$^{c}$ & \object{lm0601k18043} & 05:46:47.74 --70:38:13.5 & & yes  \\
\object{MACHO-16.5641.22}$^{c}$ & \object{lm0190l20117} & 05:14:46.20 --67:55:47.4 & HV 2379 & yes  \\
\object{MACHO-18.3325.148}$^{c}$ & \object{lm0111k3287} & 05:01:00.36 --69:03:43.2 & HV 12524 & no  \\
\object{MACHO-79.5743.15}$^{c}$ & \object{lm0091k19526} & 05:15:51.79 --69:10:08.6 & & yes  \\
\object{MACHO-80.6956.207}$^{c}$ & \object{lm0206l17380} & 05:22:57.37 --68:58:18.9 & & yes  \\
\object{MACHO-80.7559.28}$^{c}$ & \object{lm0011m13457} & 05:26:33.91 --69:07:33.4 & & yes  \\
\object{MACHO-2.5871.1759}$^{c}$ & \object{lm0196m8601} & 05:16:51.96 --68:45:16.8 & & no  \\
\object{MACHO-10.3800.35}$^{c}$ & \object{lm0102l12314} & 05:03:44.65 --69:38:11.7 & & no  \\
\object{MACHO-15.10675.10}$^{c}$ & \object{lm0603l29693} & 05:46:13.27 --71:07:40.4 & & yes  \\
\object{MACHO-78.6460.7}$^{c}$ & \object{lm0014k7770} & 05:19:55.93 --69:48:05.9 & & no \\
\object{EROS2-SMC-RCB-1}$^{d}$ & \object{sm0102l20592} & 00:37:47.11 --73:39:02.3 & RAW-21&yes\\
\object{EROS2-SMC-RCB-2}$^{d}$ & \object{sm0014k11612} & 00:48:22.96 --73:41:04.7 & RAW-476&yes  \\
\object{EROS2-SMC-RCB-3}$^{d}$ & \object{sm0067m28134} & 00:57:18.15 --72:42:35.2 & MACHO-207.16426.1662 & yes  \\
 & & & MSX-SMC-155 & \\
\object{EROS2-SMC-DYPer-1}$^{d}$ & \object{sm0077k11497} & 00:44:07.50 --72:44:16.4 & RAW-233 & yes  \\
 & & & MACHO-208.15571.60 & \\
\object{EROS2-SMC-DYPer-2}$^{d}$ & \object{sm0106m19412} & 00:40:14.72 --74:11:21.6 & [MH95]-431 & yes \\
\hline
& & RCB candidates & & \\
\object{KDM 2373}$^{e}$ & lm0105m6306 & 05:10:28.52 --69:47:04.4 & & Confirmed RCB in this article \\
\object{KDM 5651}$^{e}$ & lm0593m31181 & 05:41:23.51 --70:58:01.7 & MACHO-15.9830.5 & Oscillations and small \\
 & & & & $\sim$0.3 mag drops observed\\
\object{KDM 7101}$^{e}$ & lm0745l21924 & 06:04:05.46 --72:51:22.8 & & Confirmed RCB in this article  \\
\object{MSX-SMC-014}$^{f}$ &  & 00:46:16.33 --74:11:13.6 & monitored by OGLE-3 & Too faint to be catalogued \\
\object{MSX-LMC-755}$^{g}$ & lm0213l10369 & 05:32:56.18 --68:12:49.0 & MACHO-8.8541.68 & Large oscillation observed \\
\hline
\end{tabular}}
\medskip
\hspace*{4mm}$^{a}$ \citet{1972MNRAS.158P..11F,1996PASP..108..225C}; $^{b}$ \citet{1996ApJ...470..583A}; $^{c}$ \citet{2001ApJ...554..298A}; $^{d}$ \citet{2004AetA...424..245T}; $^{e}$ \citet{2003MNRAS.344..325M};\\ \hspace*{4mm}$^{f}$~\citet{2005ApJ...631L.147K}; $^{g}$~\citet{2006AJ....132.1890B}.\\
\end{table}